Author's response to reviews

Title: A Randomized Controlled Trial of Interventions to Enhance Patient-Physician Partnership, Patient Adherence and High Blood Pressure Control among Ethnic Minorities and Poor Persons: Study Protocol NCT00123045.

Authors:

Lisa A Cooper (lisa.cooper@jhmi.edu)
Debra L Roter (droter@jhsph.edu)
Lee R Bone (lbone@jhsph.edu)
Susan M Larson (slarson@jhsph.edu)
Edgar R Miller III (ermiller@jhmi.edu)
Michael S Barr (mbarr@mail.acponline.org)
Kathryn A Carson (karson@jhmi.edu)
David M Levine (dlevine@jhmi.edu)

Version: 4 Date: 8 February 2009

Author's response to reviews:

Dear Editors:

I have removed figure 4 (the CONSORT checklist) from the manuscript completely, made formatting changes to the tables (1 and 2), removed titles and legends from figures (1, 2, and 3) as you requested, and placed a figures legend section just after the references section of the manuscript. I have also made final minor edits to the manuscript.

Thank you very much for your attention.

Sincerely,

Lisa A. Cooper, MD